Author's response to reviews

Title: Irreversible Electroporation of Unresectable Soft Tissue Tumors with Vascular Invasion: Effective Palliation

Authors:

Robert Martin (robert.martin@louisville.edu)
Prejesh Philips (prejesh.philips@louisville.edu)
Susan Ellis (susan.ellis@louisville.edu)
David Hayes (David.hayes@gmail.com)
Sandeep Bagla (sandeep.bagla@inova.com)

Version: 7 Date: 7 July 2014

Author's response to reviews:

July 7, 2014

Annie Lyn Bravo
On behalf of The BioMed Central Team
Journal Editorial Office
BioMed Central

RE: Re-Submission Manuscript “Irreversible Electroporation of Unresectable Soft Tissue Tumors with Vascular Invasion: Effective Palliation”

Dear Editor:

Thank you for the additional review of our manuscript. We appreciate the acceptance of our manuscript. We have revised our manuscript per the Editorial request.

Please find a point by point description of those changes.

In Response to Editorial comments:

1. The ethical statement has been inserted.
2. The manuscript has been re-reviewed for grammatical errors and corrected
Thank you for allowing us to resubmit our manuscript

Sincerely,

Robert C. G. Martin, II, MD, PhD, FACS
Sam and Lolita Weakley Chair in Surgical Oncology
Professor of Surgery
Director, Division of Surgical Oncology
Director of the Upper GI and HPB Multi-Disciplinary Clinic
Academic Advisory Dean